Author's response to reviews

Title: Is Bioelectrical Impedance accurate for use in large epidemiological studies?

Authors:

   Mahshid Dehghan (mahshid@ccc.mcmaster.ca)
   Anwar T Merchant (anwar_merchant@yahoo.com)

Version: 4 Date: 1 September 2008

Author's response to reviews: see over
Dear Editor,

Re: Is Bioelectrical Impedance accurate for use in large epidemiological studies?

On behalf of my co-author, I would like to thank you for accepting the above manuscript for publication in the Nutrition Journal. Regarding the major change that suggested by editorial team I would like to bring to your attention that our manuscript is a review of published articles and ethics approval and consent form were not needed. We have implemented all minor changes. We appreciate the opportunity to improve the quality of the manuscript.

Sincerely,
Mahshid Dehghan Ph.D, RNutr (UK)
Department of Medicine, McMaster University
Population Health Research Institute
Hamilton General Hospital
237 Barton Street East
Tel: 905-527-4322 Ext: 44457
Fax: 905-527-9642